Dear editor,

Please find attached our revised editorial. We thank the reviewer for his comments. We included in the manuscript the reference suggested. We added:

Conductance disturbances and the need for permanent pacemaker implantation post-TAVR are multifactorial and have an important clinical and economic impact (price of the pacemaker implantation, higher length of stay) [5].

Indeed, not only does the baseline ECG influence the post-procedural risk of permanent pacemaker need (particularly the presence of a right bundle branch block [5]),

We hope that the manuscript will be of interest to your readers.

Stephane Noble and Karim Bendjelid

Signature numérique de BENDJELID
Date : 2024.01.30
16:31:13 +01'00'